\nocite{Harlan.sep.32.25}
\nocite{Claerbout.sep.32.37}
\nocite{Harlan.sep.32.79}
\nocite{Harlan.sep.35.81}
\nocite{Harlan.sep.35.107}
\nocite{Harlan.sep.35.181}
\nocite{Harlan.sep.37.121}
\nocite{Harlan.sep.38.315}
\nocite{Harlan.sep.41.283}
\nocite{Harlan.sep.41.405}
\nocite{Harlan.sep.41.431}
\nocite{Harlan.sep.42.159}
\nocite{Harlan.sep.42.173}
\nocite{Harlan.sep.44.85}
\nocite{Harlan.sep.47.0}
\nocite{Harlan.sep.47.57}
\nocite{Harlan.sep.47.95}
\nocite{Harlan.sep.47.127}
\nocite{Harlan.sep.47.129}
\nocite{Harlan.sep.47.131}
\nocite{Harlan.sep.47.135}
\nocite{Harlan.sep.47.139}
\nocite{Harlan.sep.47.143}
\nocite{Harlan.sep.47.145}
\nocite{Harlan.sep.47.147}
\nocite{Harlan.sep.48.127}
\nocite{Harlan.sepphd.47}
\nocite{Harlan.sep.47}
\nocite{Harlan.sep.57.313}
\nocite{Harlan.sep.60.215}


\title{Report bibliography}
This is my bibliography of articles
in the progress reports of the Stanford Exploration Project
written from 1974 to the present.
There are  29 
articles.
{\footnotesize
\putbib[SEP]
}
%   $Id: autobiblio.tex,v 1.1 1999/03/28 09:06:52 harlan Exp $
%   @(#)autobiblio.tex	1.1     08 Jul 1996 
%   $Id: autobiblio.tex,v 1.1 1999/03/28 09:06:52 harlan Exp $
%   $Id: autobiblio.tex,v 1.1 1999/03/28 09:06:52 harlan Exp $
